# RESEARCH



# Invisible walls? Stigma-related perceptions are associated with reduced help-seeking intentions for disordered eating in men



Martin S. Lehe<sup>1,2\*</sup>, Georg Halbeisen<sup>1</sup>, Sabine Steins-Loeber<sup>2</sup> and Georgios Paslakis<sup>1</sup>

# Abstract

**Background** Eating disorders (EDs) are increasingly prevalent in men, but men remain underrepresented across many ED-specific treatment settings. Based on the idea that persistent stereotypes, prejudice and discrimination, i.e., stigma against men with EDs, could impede help-seeking behaviors, the present study investigated whether stigma-related perceptions in men are associated with reduced help-seeking intentions for a broad range of disordered eating symptoms.

**Methods** N = 132 adult men participated in a cross-sectional online survey and completed questionnaires on ED psychopathology, muscle dysmorphia, orthorexic eating, stigma-related perceptions of EDs in men, and help-seeking intentions.

**Results** Moderator analyses showed that higher stigma-related perceptions were associated with reduced helpseeking intentions in response to increased ED symptom severity. However, this was only the case for traditionally "feminized" ED symptoms (related to thin-body ideals), but not for help-seeking with regard to muscularity-oriented, orthorexic, or avoidant/restrictive disordered eating.

**Conclusions** Stigma may reduce help-seeking intentions with regard to "feminized" ED symptoms. The present findings suggest that perceptions of EDs as "women's diseases" were associated with reduced help-seeking in men. Stigma towards men with EDs could thus be a possible barrier to help-seeking in men, highlighting the relevance of stigma-reducing interventions in clinical and community settings.

# **Plain English Summary**

Eating disorders are associated with many burdens for those affected, their relatives, and the healthcare system. In recent years, there has been a significant increase in the incidence of eating disorders—particularly among men. Nevertheless, data shows that men are less likely than women to receive adequate treatment. Drawing on reports of individuals with lived experience, several barriers to treatment are commonly discussed in the literature. One of these is stigma, commonly present in men with eating disorders. This study empirically supports the notion that stigmarelated perceptions may be a barrier to seeking treatment for men with disordered eating. This may help to develop interventions on different levels to break down barriers to treatment for men.

Keywords Eating disorders, Male, Men, Stigma, Gender, Treatment barrier, Access to care, Help seeking

\*Correspondence: Martin S. Lehe martin.lehe@rub.de Full list of author information is available at the end of the article



© The Author(s) 2024. **Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/.

# Background

Eating disorders (EDs) are severe psychiatric conditions characterized by body image concerns, eating disturbances, and weight-control behaviors that impose substantial levels of individual [1, 2], caregiver [3, 4], and economic burden [5]. Although considered one of "the most gendered of all disorders" [6], EDs affect a socially diverse range of individuals [7, 8], including adult men. In fact, ED prevalence in men increased by 22% over the past three decades compared to a 12% increase in women [9]. Women remain in the majority, but current lifetime estimates of 2.2% ED prevalence in men and 8.4% in women suggest that men could account for every fourth to fifth clinical case [8, 10].

However, despite seminal case reports of men with EDs [11–13], men remain underrepresented across many ED treatment settings [14]. This may be because men's symptoms may go unnoticed due to gender differences in body image and presentation of disordered eating [15-17]. For example, disordered eating in men has been associated with orthorexia symptoms, exercise dependence, the drive for muscularity, and muscle dysmorphia [18, 19]. In some settings, lesser-known ED diagnoses may also be more common in boys and men (e.g., avoidant/restrictive food intake disorder, ARFID, in pediatric gastroenterology [20]. Still, one of the most prominent reasons for low ED treatment rates in men may be persistent stereotypes, prejudice, and discrimination, that is, stigma against men with EDs and subsequently delayed or thwarted helpseeking behaviors [21].

Adult men presumably delay seeking help for healthrelated issues, particularly those pertaining to mental health, due to the conflict between help-seeking behavior and ideals of masculinity and traditional gender roles [22, 23]. Traditional gender roles require men to be strong and self-reliant, and to avoid showing weakness and vulnerability [24]. These gender roles are socially reinforced, such that individuals experience reward for adhering and social backlash, i.e., social and economic penalties, for not adhering to societal expectations [25], incentivizing inaction regarding the disclosure of problems and seeking help. Furthermore, manhood may be perceived as a status that is precarious and potentially lost by displaying "feminine" behaviors or behaviors that fail to demonstrate masculinity [26]. Thus, seeking help could not only conflict with masculinity ideals and traditional gender roles, but actively threaten a man's perceived social status [27]. Consequently, seeking help for having disordered eating, a "feminized" mental health problem that violates gender ideology, appears particularly problematic and might face significant social and individual barriers.

Indeed, two recent systematic reviews suggest that men with disordered eating encounter unique challenges in accessing care. Richardson and Paslakis [28] summarized qualitative studies on men's lived experiences, revealing, for example, men with EDs having trouble finding specialized treatments, physicians not taking symptoms in men with EDs seriously, and men delaying help-seeking due to the internalized notion that EDs are a "women's disease". Including quantitative studies and focusing specifically on barriers to help-seeking in men, Bomben et al. [29] similarly concluded that perceiving services as "feminized" and EDs as a "female illness" have adverse effects on men. Specifically, five cross-sectional surveys showed that men and women were equally likely to recognize ED symptoms [30] but that men with a diagnosed [31] or self-suspected ED [32] expected more negative consequences from seeking psychological help in general compared to women. Such generalized negative expectations further emerged as a stronger predictor of lower help-seeking intentions in men with disordered eating compared to women [33, 34], which Bomben et al. [29] attributed to a feminized construction of EDs actively thwarting men's chances of recovery. After all, admitting one's symptoms [35] and then intending to seek help [36] are initial steps and significant predictors for recovery and ED treatment outcomes.

While the overall findings are consistent with a critical association between delayed help-seeking and stigma-related perceptions of EDs in men (e.g., "EDs are a women's disease."; "There aren't any services for men."; "Physicians will not take men's symptoms seriously."), it is important to note that the previous studies rarely tested the impeding effect of these stigma-related perceptions directly. The cited studies almost exclusively measured men's perceptions towards seeking psychological help in general. The pronounced rates in men [31] and associations with disordered eating [32] or general help-seeking intentions [33] thus only indirectly support the notion that it is stigma-related perceptions towards men with ED that are associated with reduced help-seeking intentions. None of the cited studies actually measured stigma-related perceptions of EDs in men, which could be distinct from generalized negative expectations of seeking help [37]. In addition, previous studies with men mainly included "femalecentric" disordered eating assessment (e.g., the Eating Disorder Examination-Questionnaire; [38, 39]) and did not consider stereotypically "male" body image concerns and disordered eating presentation [6, 40]. It, therefore, remains an open question whether reduced help-seeking intentions for other disordered eating symptoms would be similarly associated with stigmarelated perceptions of EDs in men or if the effect, as is currently assumed [29], specifically concerns "feminized" disordered eating symptoms.

To address the open questions and extend upon previous knowledge on how stigmatization could affect men [41], we investigated the effect of stigma-related perceptions of EDs in men on men's help-seeking intentions for disordered eating symptoms. Specifically, we collected data on men's disordered eating symptoms, their stigma-related perceptions of EDs in men, and their help-seeking intentions in a cross-sectional survey. We generally assumed increased help-seeking intentions in individuals admitting to more disordered eating symptoms [35]. However, based on the idea that a feminized construction of EDs is associated with reduced helpseeking behaviors in men [28, 29], we further predicted that more pronounced stigma-related perceptions of EDs in men would moderate, i.e., reduce the relation between symptom severity and help-seeking intentions. Because we collected data on a broader spectrum of disordered eating behaviors, we further explored whether this association specifically occurred with "feminized" disordered eating symptoms (i.e., those related to thin body ideals and weight control behaviors) or also disordered eating symptoms potentially more relevant-and less stigmatized—in men (e.g., muscularity concerns, or ARFID [6, 15]).

# Materials and methods Participants and design

The present study included data of N=132 adult men  $(M_{age} = 33.4, SD_{age} = 12.4, age range: 18-68 years)$  as a subset of a cross-sectional online survey [42]. The initial study comprised a sample of 158 participants in total, from whom 146 were men meeting the inclusion criteria. Participants were recruited from the local university campus, in online social networks, and among acquaintances between May 2022 and June 2023. For the present analysis, we included the subsample of adult men who answered 'no' to having ever received a diagnosis for an eating disorder, given that we were interested in predicting help-seeking intentions unaltered by prior ED-related help-seeking experiences. An a priori power analysis was conducted using G\*Power version 3.1.9.7 [43] for sample size estimation. This was based on the assumption of a small-to-medium-sized effect ( $\rho = 0.10$ ) according to Cohen's criteria [44], which had been observed in previous research [45]. With a significance criterion of  $\alpha = 0.05$ and power = 0.80, the minimum sample size needed with this effect size was N=100 for moderation analyses with two interaction effects. Thus, the obtained sample size was adequate to test the study hypotheses.

The survey was approved by the Ethics Committee of the Ruhr-University Bochum's Medical Faculty at Campus East-Westphalia (AZ 2022–910, April 21st, 2022), prospectively registered at https://aspredicted.org/5T3\_ NH5, and conducted in accordance with the Declaration of Helsinki. All participants gave informed consent. Participants received no compensation. We report all measures and exclusions. Data and materials can be obtained from the corresponding author upon request.

## Measures and procedure

The online survey was hosted on our webserver and implemented using jsPsych [46]. The study was advertised as a survey on improving the detection of disordered eating in men. Upon accessing the study's website, the study information and consent forms were presented. We explained that the survey included the assessment of disordered eating and other symptoms in men and that participants could receive feedback on their risk for an eating disorder. Participants first provided sociodemographic information (including age, weight, height, regular medications, gender, sexual orientation, German language proficiency, migration background, years of education, marital status, living circumstances, ED history and treatment, and previous study participation). The latter question served as a quality check to exclude records from repeated participation. Participants next completed an ED risk assessment (EAT-8 [47]; for further details, see [42]), and, in randomized order, disordered eating symptoms, help-seeking intentions, and stigmarelated perceptions of EDs in men. The survey took between 15 to 20 min to complete.

#### Disordered eating symptoms

Similar to previous studies [31, 32], we used the Eating Disorder Examination-Questionnaire (EDE-Q; [48, 49]) to assess cognitive and behavioral ED symptoms within the preceding 28 days. The EDE-Q encompasses four subscales, namely "(Dietary) Restraint", "Eating Concern", "Shape Concern", and "Weight Concern" utilizing 22 attitudinal items rated on a 7-point scale (ranging from 0, *never*, to 6, *every day*). Six additional open-ended items assessed overeating episodes, binge episodes, binge days, and purging behaviors but were not used in scale construction. Since earlier investigations did not endorse the proposed factor structure of the EDE-Q in men [39], we opted not to consider the subscale scores. Instead, we used the global mean score across attitudinal items (Cronbach's  $\alpha$ =0.94, McDonald's  $\omega$ =0.94).

Extending upon previous studies, we also assessed additional disordered eating and associated symptoms. The Muscle Dysmorphic Disorder Inventory (MDDI; [50]) comprised 13 items, rated on a 5-point scale (from 1, *never*, to 5, *always*), assessing disturbances and behaviors associated with muscularity-related body image (e.g., the belief that one's body is too skinny, hating one's body, and experiencing depressed mood when not engaging in

Item	М	SD
1. When men eat too little, are obese or exercise excessively, it is usually not due to an eating disorder	2.45	0.69
2. Eating disorders are women's diseases	1.48	0.69
3. It is better for a man to not admit having an eating disorder	1.33	0.66
4. Men with eating disorders have a harder time finding treatment than women	2.26	0.79
5. Doctors take eating disorders less seriously in men than in women	2.39	0.92
6. There are hardly any specialized treatment services for men with eating disorders	2.39	0.90
7. Eating disorders are less treatable for men than for women	1.82	0.66
Total score	2.02	0.45

Table 1 Items and descriptive characteristics of the self-developed stigma-related perceptions of eating disorders in men measure

N = 132. Items were rated on a 4-point scale from 1 (completely disagree) to 4 (completely agree) and aggregated such that higher scores indicated higher stigmarelated perceptions of eating disorders in men

exercise; Cronbach's  $\alpha = 0.82$ , McDonald's  $\omega = 0.76$ ). The Duesseldorf Orthorexia Scale (DOS; [51]) assessed the preoccupation with healthy eating and related behaviors in the past week using ten statements (e.g., "I prioritize healthy eating over pleasure") rated on a 4-point scale ranging from 1, this does not apply to me, to 4, this applies to me (Cronbach's  $\alpha = 0.82$ , McDonald's  $\omega = 0.80$ ). Finally, the Eating Disorders in Youth-Questionnaire (EDY-Q; [52]), adapted and validated for adults in Germany [53], examined ARFID symptoms through 14 questions encompassing food avoidance, selective eating, functional dysphagia, and issues related to underweight on a 7-point scale ranging from 0, never true, to 6, always *true* (Cronbach's  $\alpha = 0.66$ , McDonald's  $\omega = 0.61$ ). The total score is represented by the mean score of items 1 to 5 and 8 to 12, as per the convention of the instrument.

#### Help-seeking intentions

We measured help-seeking intentions with an item from the validated German version of the Stages of Change Questionnaire for Eating Disorders (SOCQ-ED; [54]). The SOCQ-ED is based on the transtheoretical model of change [55] and gauges a participant's motivation to change specific ED symptoms and the overall intention to seek treatment for disordered eating, with a 7-point scale reflecting different intentional states (no need for therapy, precontemplation, contemplation, preparation, action, maintenance, termination; coded from 0 to 6). Because the overall item directly measured help-seeking intentions, we included it as the primary outcome variable in the present study.

## Stigma-related perceptions of disordered eating in men

In the absence of a pre-existing (German) measure on stigma-related perceptions of EDs in men, we developed a set of seven items based on the lived experiences of men with EDs identified by Richardson and Paslakis [28]. Specifically, through discussion and drawing upon our clinical expertise, we formulated items that addressed perceived structural, public, and internalized aspects of stigmatization (e.g., "There are hardly any specialized treatment services for men with eating disorders.", "Doctors take eating disorders less seriously in men than in women.", "It is better for a man to not admit having an eating disorder"). The items were rated on a 4-point scale from 1, *completely disagree*, to 4, *completely agree*, and aggregated such that higher scores indicated higher perceived stigma. Table 1 shows the complete item list. The mean score had approximately acceptable internal consistency (Cronbach's  $\alpha = 0.69$ , McDonald's  $\omega = 0.71$ ).

#### Data aggregation and analysis

The disordered eating questionnaires were aggregated according to their convention. Variable values are reported as means (M) and standard deviations (SD). We screened for multivariate outliers across questionnaire scores using Mahalanobis distance with a criterion of p < 0.001. The internal consistency of the questionnaire scales was assessed using Cronbach's a and McDonald's  $\omega$  coefficients. The items on stigma-related perceptions of EDs in men were submitted to a single-factor confirmatory factor analysis (CFA) before aggregation. The CFA was conducted in R version 4.3.1 [56] using package lavaan version 0.6-17 [57] with maximum likelihood estimation and standardization of both observed and latent variable covariances. Model fit was assessed based on common fit indices, i.e., Comparative Fit Index (CFI), Root Mean Squared Error of Approximation (RMSEA), Standardized Root Mean Square Residual (SRMR), and Tucker-Lewis Index (TLI). In accordance with previous literature and recommendations [58, 59], we considered the model to fit the data well when CFI and  $TLI \ge 0.95$ , RMSEA < 0.06 and SRMR < 0.08. For the main analyses we used Pearson correlations to examine bivariate associations. Moderator analyses, with disordered eating as the predictor, stigma-related perceptions of EDs in men as

the moderator, and help-seeking intentions as the criterion, were performed using the PROCESS macro version 4.2 [60] for IBM SPSS 29 [61], with bootstrapped (5,000 samples) bias-corrected 95% confidence interval (CI) and heteroscedasticity correction (HC3). Predictor and moderator variables were mean-centered. Simple slopes were evaluated at the moderator mean (M) and  $M \pm 1$  SD and, more granular, using the Johnson-Neyman technique [62]. The significance level for all analyses was set at  $p \leq 0.05$ . We used case-wise exclusion across analyses for missing data (1.9%). Plots for moderations were created using the ggplot2 package version 3.4.3 [63] in R version 4.3.1 [56] and an online visualization tool [64].

### Results

# Sample characteristics

Table 2 summarizes the sample sociodemographic characteristics, disordered eating symptoms, and help-seeking intentions. We did not detect multivariate outliers. Consistent with the selection of men without EDs, the overall symptom endorsement and help-seeking intentions remained low. The EDE-Q scores correlated positively with the MDDI and DOS, but not with the EDY-Q (Table 3). Age was negatively correlated with muscle dysmorphic symptoms (MDDI) but not with the other disordered eating scores. Stigma-related perceptions of EDs in men were pronounced in younger men and those who were single; help-seeking intentions did not vary by sociodemographic features (Table 4).

#### Men's ED stigma and help-seeking intentions

The CFA showed that a single latent factor fits the data well (CFI=0.99, RMSEA=0.02, SRMR=0.05, TLI=0.99; for full results, see Appendix). Because the items were designed to capture the full range of previously reported stigma-related experiences [28], we aggregated all items into a single mean score for further analyses to preserve the measure's construct validity, despite individual low-loading items.<sup>1</sup> The overall endorsement of stigma-related perceptions of EDs in men fell between slight disagreement and slight agreement, M=2.02, SD=0.45 (see Table 1). The moderator analysis with EDE-Q scores as predictor X,

Table 2	Sample	characteristics,	disordered	eating	symptoms,
and help	-seeking	intentions			

Parameter	Total	Min	Max
n	132		
Age (years)	33.4 (12.4)	18.0	68.0
BMI (kg/m <sup>2</sup> )	26.2 (5.8)	17.1	52.3
Gender			
Male	131 (99.2%)		
Diverse	1 (0.8%)		
Sexual orientation			
Women	112 (84.8%)		
Men	16 (12.1%)		
Women and men	3 (2.3%)		
Other	1 (0.8%)		
German language proficiency			
First language	128 (97.0%)		
Fluent	4 (3.0%)		
Educational attainment			
Less than 12 years	22 (16.7%)		
12 years or more	110 (83.3%)		
Migration background			
Yes	14 (10.6%)		
No	118 (89.4%)		
Marital status			
Married	37 (28.0%)		
Single	91 (68.9%)		
Divorced	4 (3.0%)		
Living situation			
Alone	39 (29.5%)		
With others	93 (70.5%)		
EDE-Q global mean	1.34 (1.16)	0.00	4.73
DOS sum score	17.23 (4.66)	10	32
EDY-Q <sup>a</sup> mean	1.21 (0.76)	0.00	4.20
MDDI sum score	26.47 (7.71)	13	53
Help-Seeking Intentions (SOCQ-ED)	0.49 (0.98)	0.00	6.00

Values are shown in n (%) or means and standard deviations (in brackets). Min = Minimum Value; Max = Maximum Value. EDE-Q = Eating Disorder Examination-Questionnaire; DOS = Duesseldorf Orthorexia Scale; EDY-Q = Eating Disorders in Youth-Questionnaire; MDDI = Muscle Dysmorphic Disorder Inventory; SOCQ-ED = Stages of Change Questionnaire for Eating Disorders

<sup>a</sup> N = 118 due to missing responses

stigma-related perceptions as moderator *W*, and helpseeking intentions as criterion *Y*, was significant, *F*(3, 128) = 16.71, p < 0.001, R2 = 0.41. As expected, increased EDE-Q scores were associated with increased helpseeking intentions,  $b_{X \rightarrow Y} = 0.53$ , SE = 0.08, p < 0.001, 95% CI [0.39; 0.68]. Stigma-related perceptions were not directly associated with help-seeking intentions,  $b_{W \rightarrow Y} = -0.37$ , SE = 0.22, p = 0.09, 95% CI [-0.78; 0.02]. However, stigma-related perceptions modulated the

<sup>&</sup>lt;sup>1</sup> Despite the CFA showing adequate fit, the internal consistency of the items was suboptimal (Cronbach's  $\alpha$ =.69, McDonald's  $\omega$ =.71). We therefore re-analyzed our data based on the splitting of items suggested by an exploratory factor analysis (EFA; see Appendix A). This re-analysis more specifically linked a four-item factor of stigma-related perceptions of ED treatment to help-seeking behavior rather than a single-item factor of more general gendered ED stereotypes. However, because the overall results and conclusions remained comparable when splitting the items into two factors or when only including the four-item factor, and due to the difficulty of interpreting single-item factors, we adhered to reporting the intended analyses using the overall mean.

#### Table 3 Bivariate variable correlations

Variable	N	М	SD	1	2	3	4	5	6	7
1. Age	132	33.36	12.39	1						
2. BMI	132	26.20	5.83	0.27**	1					
3. EDE-Q	132	1.34	1.16	0.07	0.44**	1				
4. DOS	132	17.23	4.66	-0.07	0.14	0.52**	1			
5. EDY-Q <sup>a</sup>	118	1.21	0.76	- 0.03	-0.14	0.13	-0.03	1		
6. MDDI	132	26.47	7.71	-0.24**	0.07	0.62**	0.56**	0.32**	1	
7. Stigma	132	2.02	0.45	-0.19*	-0.08	0.11	0.25**	0.09	0.37**	1

BMI = Body Mass Index; EDE-Q = Eating Disorder Examination-Questionnaire; DOS = Duesseldorf Orthorexia Scale; EDY-Q = Eating Disorders in Youth-Questionnaire; MDDI = Muscle Dysmorphic Disorder Inventory; Stigma = Stigma-related perceptions of eating disorders in men

<sup>a</sup> N = 118 due to missing responses

\* *p* < 0.05. \*\**p* < 0.01

 Table 4
 Perceived stigma and help-seeking intentions by sociodemographic characteristics

Educational attainment	<12 years (n = 22)	>12 years (n = 110)	t	p	d
Stigma	1.94 (0.45)	2.03 (0.45)	-0.95	0.352	-0.22
SOCQ-ED	0.45 (0.86)	0.49 (1.00)	-0.16	0.874	-0.04
Marital status	Single ( <i>n</i> =91)	Married (n = 37)	t	p	d
Stigma	2.10 (0.43)	1.90 (0.40)	2.33	0.021	0.46
SOCQ-ED	0.49 (0.86)	0.24 (0.60)	1.89	0.062	0.32
Living situation	Alone ( <i>n</i> = 39)	With others (n=93)	t	p	d
Stigma	2.01 (0.47)	2.02 (0.44)	- 0.04	0.965	-0.01
SOCQ-ED	0.44 (0.94)	0.51 (1.00)	-0.37	0.711	-0.07
Sexual orientation	Non-heterosexual (n = 19)	Heterosexual (n = 112)	t	p	d
Stigma	2.17 (0.46)	1.99 (0.44)	1.61	0.111	0.40
SOCQ-ED	0.63 (1.12)	0.46 (0.96)	0.69	0.493	0.17

Values show means and standard deviations (in parentheses). Stigma = Stigma-related perceptions of eating disorders in men; SOCQ-ED = Stages of Change Questionnaire for Eating Disorders

effect of symptom severity on help-seeking intentions,  $b_{X^*W \to Y} = -0.34$ , SE = 0.17, p = 0.04, 95% CI [-0.66; -0.02],  $R^2_{\text{change}} = 0.04$ . Consistent with our prediction, simple slope analysis (see Fig. 1) showed that the effect of symptom severity on help-seeking intentions decreased with increasing levels of stigma-related perceptions,  $b_{-1SD} = 0.68$ ,  $b_M = 0.53$ ,  $b_{+1SD} = 0.38$ , although the effect remained significant at the evaluated mean  $\pm 1$  SD levels, all ps < 0.001. However, the more granular Johnson-Neyman analysis showed that EDE-Q scores were only associated with help-seeking intentions until a stigma-related perceptions score of 2.82 (i.e., a tendency toward slight agreement), and became non-significant afterward, consistent with the idea that stigma-related perceptions are associated with reduced help-seeking for EDs in men (see Fig. 2). The results remained unaltered when including sociodemographic factors associated with reduced help-seeking such as low educational status, young age, and never-married status as covariates [65].

We further explored moderations by stigma-related perceptions for the effects of other disordered eating symptoms (see Table 5). The model with muscle dysmorphia (MDDI) scores as predictor was significant, F(3,128) = 6.51, *p* < 0.001,  $R^2$  = 0.16. Higher MDDI scores and lower stigma-related perceptions predicted more helpseeking intentions,  $b_{\chi \to \gamma} = 0.05$ , *SE* = 0.01, *p* < 0.001, 95% CI [0.03; 0.08] and  $b_{W \to Y} = -0.65$ , SE = 0.29, p = 0.03, 95% CI [-1.22; -0.13], respectively. However, stigma-related perceptions did not modulate the effect of the MDDI,  $b_{X^*W \to Y} = -0.01$ , SE = 0.02, p = 0.56, 95% CI [-0.05; 0.04]. The model with orthorexic symptoms (DOS) as predictor was also significant, F(3, 128) = 3.38, p = 0.020,  $R^2 = 0.13$ . More pronounced DOS scores predicted increased helpseeking intentions,  $b_{X \rightarrow Y} = 0.07$ , SE = 0.02, p = 0.009, 95% CI [0.03; 0.12], but stigma-related perceptions had



Fig. 1 Moderating Effect of Stigma-Related Perceptions of Eating Disorders in Men on the Relationship Between ED Symptom Severity and Help-Seeking Intentions. ED = eating disorder. EDE-Q = Eating Disorder Examination-Questionnaire

neither a direct influence nor did it modulate the effect of the DOS,  $b_{W\rightarrow Y} = -0.49$ , SE = 0.25, p = 0.05, 95% CI [-1.00; -0.02] and  $b_{X^*W\rightarrow Y} = -0.03$ , SE = 0.05, p = 0.57, 95% CI [-0.13; 0.07], respectively. Finally, EDY-Q ARFID scores were unrelated to help-seeking intentions and were not modulated by stigma-related perceptions, F(3, 114) = 1.03, p = 0.382,  $R^2 = 0.05$ . Thus, we did not obtain any evidence that stigma-related perceptions of EDs in men modulated the effects on help-seeking intentions of disordered eating symptoms besides those assessed by the EDE-Q.

# Discussion

Despite an increasing number of cases, men remain underrepresented in settings for the treatment of disordered eating. Previous studies found that men with disordered eating report more stigma towards seeking psychological help in general, which has been attributed to a feminized construction of EDs impeding helpseeking behavior [28, 29]. Here, we investigated whether stigma-related perceptions of EDs in men moderated the association between symptom severity and help-seeking intentions. Corroborating previous assumptions, we found that higher levels of disordered eating symptoms, as assessed by a standard measure of ED psychopathology (EDE-Q), predicted men's intention to seek help and that the effect diminished with increasing levels of endorsed stigma-related perceptions of EDs in men. Specifically, disordered eating symptoms ceased to be associated with help-seeking intentions with even slight levels of stigmarelated perceptions of EDs (2.82 on a scale of 1 to 4), which suggests that perceived stigmatization could possibly prevent men from seeking help for self-recognized symptoms.

Notably, the present findings provide initial evidence for the specificity of stigma-related barriers to help-seeking. Previous studies associated disordered eating in men with a broad range of symptoms, including muscularityoriented, orthorexic, and avoidant/restrictive forms of eating behaviors [15–17]. We therefore assessed these behaviors, which mostly correlated positively with more traditional, "feminized" disordered eating symptoms (i.e., thinness, weight concern as assessed by EDE-Q). We also found that more pronounced muscle dysmorphia and orthorexic symptoms similarly predicted helpseeking intentions. ARFID symptoms were unrelated to help-seeking intentions and the association was also not modulated by stigma-related perceptions, although the low reliability of the measure found here and in other studies [53] limits interpretations. However, stigma only



Fig. 2 Johnson-Neyman Significance Regions for the Conditional Effects of Symptom Severity on Help-Seeking Intentions at Different Values of Stigma-Related Perceptions of Eating Disorders in Men. The effect is significant at stigma-related perceptions values below the dashed, vertical line (2.82 on a scale of 1 to 4). ED=eating disorder. EDE-Q=Eating Disorder Examination-Questionnaire

modulated the effect of the traditional, but not of the other disordered eating symptoms, for predicting helpseeking intentions, which directly supports that a "feminized" construction of EDs, and not of disordered eating per se, may impede access to care [29]. While the overall effects of a broad range of symptoms on help-seeking intentions highlight that men may benefit from more inclusive assessments in clinical practice, the potential specificity of the stigma-related moderation reiterates the importance of continuing efforts to tackle stigmatized perceptions of EDs as a "women's disease" [66].

Although the CFA conducted showed that the different aspects assessed by the stigma-related perceptions item set could be summarized as a single construct, as the results were discussed previously, the supplementary exploratory factor analysis (EFA, see Appendix) suggested splitting the item set into two factors. This suggests that men's stigma-related perceptions of ED treatment may be more specifically linked and therefore more relevant to help-seeking behavior than a singleitem factor of more general gendered ED stereotypes. However, both factors represent men's perceptions of EDs in men compared to women, supporting the notion that it may yet be gendered perceptions and constructions of EDs which are a specific barrier to treatment for men. Furthermore, given the weak and unstable nature of factors with fewer than three items [67], further interpretation should be undertaken with considerable caution.

#### Limitations

Despite the data supporting several of our assumptions and corroborating previous research, it is important to note some limitations. First, we included a general population sample and excluded individuals with a history of

Relationship	Model sta	tistics		Path statistics				
	F	df1, df2	p	b	SE	p	95% CI [ <i>LL, UL</i> ]	
EDE-Q	16.71	3, 128	< 0.001**					
X→Y				0.53	0.08	< 0.001**	0.39; 0.68	
W→Y				-0.37	0.22	0.091	-0.78; 0.02	
X*W→Y				-0.34	0.17	0.040*	-0.66; -0.02	
MDDI	6.51	3, 128	< 0.001**					
X→Y				0.05	0.01	< 0.001**	0.03; 0.08	
W→Y				- 0.65	0.29	0.026*	-1.22; -0.13	
X*W→Y				-0.01	0.02	0.561	-0.05; 0.04	
DOS	3.38	3, 128	0.020*					
X→Y				0.07	0.02	0.009*	0.03; 0.12	
W→Y				-0.49	0.25	0.053	-1.00; -0.02	
X*W→Y				-0.03	0.05	0.567	-0.13; 0.07	
EDY-Q	1.03	3, 114	0.382					
X→Y				0.02	0.12	0.844	-0.23; 0.24	
W→Y				-0.50	0.30	0.119	- 1.13; 0.06	
X*W→Y				-0.18	0.40	0.681	-0.97; 0.65	

Table 5 Moderation analyses for stigma-related perceptions

Values were computed with PROCESS version 4.2 model 1 using bootstrapping procedure with 5000 iterations, heteroscedasticity correction (HC3), and mean centering of predictor and moderator variables. X = respective symptom severity as independent variable; Y = intention to seek help as dependent variable; W = mean score of stigma-related perceptions of eating disorders in men items as moderator. EDE-Q = Eating Disorder Examination-Questionnaire; MDDI = Muscle Dysmorphic Disorder Inventory; DOS = Duesseldorf Orthorexia Scale; EDY-Q = Eating Disorders in Youth-Questionnaire; CI = confidence interval; LL = lower limit; UL = upper limit \*p < 0.05. \*\*p < 0.01

EDs to gauge effects on help-seeking intentions in the absence of prior ED-related help-seeking experiences. However, men with a diagnosed ED are potentially more affected by ED-related stigma and could show different levels of stigma-related perceptions. Indeed, the observed overall low endorsement levels of stigma-related perceptions and disordered eating symptoms may not be representative of a clinical sample. A full assessment of the association with stigma-related perceptions, therefore, requires replicating the present findings in men with diagnosed EDs. Second, we examined the effects of stigma cross-sectionally, which limits inferences concerning the causality of observed associations. Future studies should ideally include longitudinal data. Third, our study design, by using gender-specific wording and assessing gender at the beginning of the survey, may have contributed to an increased salience of gender among respondents and thus may have amplified the influence of stigma-related perceptions. We also advertised the survey as developing an online screening for EDs in men. Although transparent and anonymous, it cannot be ruled out that this may have deterred some people from participating. Relatedly, recruitment via social networks could lead to an overestimation or distortion due to snowballing effects, although we have minimized this risk by implementing a broader recruitment strategy. Although we screened for multivariate outliers and repeated participations, we did not embed attention check items into the survey, which may be considered a limitation in terms of data quality for data collected online. Fourth, we must recognize that stigma is a complex phenomenon. While we have focused here on the potential effects of a feminized construction of EDs, other facets such as weight stigma remain unexamined in this study. Future studies should therefore examine the multi-dimensionality of stigma as well as systematically assess potentially related and underlying constructs and processes, such as the precariousness of manhood or gender typing. Aggravatingly, there is a lack of consensus regarding terminology and theory within the field of stigma research [68], which should also be the focus of future efforts, as this could guide the development of more sophisticated measurement instruments and further advances in the field. For example, our items were based on a comprehensive systematic review of experiences of men with EDs [28], but future studies should aim to improve consistency while maintaining content validity. Fifth, we focused on one specific aspect of ED-related perceived stigma but did not assess its interplay with further stigma-related barriers to treatment (e.g., the likelihood of receiving a diagnosis or the general availability of specialized services; [69]). Help-seeking is a complex, multi-stepped process of evaluating one's symptoms, gaining awareness of and accepting the problem, determining the cost and

benefits of seeking help, and finally seeking and utilizing specialized services [70]. Thus, stigma may impede helpseeking intentions and behaviors in various yet unexplored ways. While using a single-item measure to assess help-seeking is efficient, it does not capture help-seeking behavior in its full complexity. Therefore, future studies need to consider utilizing more nuanced and multifaceted measures of help-seeking. Finally, complexity and precision could be added by assessing the effect of stigma through an intersectional lens, as male gender is only one of the various aspects of diversity that make up a person's identity and thus may become subject to stigmatization.

## Conclusions

EDs are an often overlooked but increasingly relevant health problem in men. Perceived stigma towards men with EDs reduces the relation between "traditional" ED symptom severity and men's intention to seek help, and thus potentially acts as a barrier to treatment for men. In this respect, the present study may be seen as an initial empirical indication of potential adverse effects of a feminized construction of EDs on help-seeking in men with disordered eating behavior, although the absence of this mechanism in women has yet to be demonstrated to confirm the assumed gender specificity. Understanding how stigma limits access to care may help to inform the development of stigma-reducing interventions on individual, societal, and structural levels [41]. It may ultimately help tear down the invisible walls towards adequate, timely, and appropriate treatment.

# Appendix

#### Confirmatory factor analysis: full results

The items on stigma-related perceptions of EDs in men were submitted to a single-factor confirmatory factor analysis (CFA) before aggregation. The CFA was conducted in R version 4.3.1 [56] using package lavaan version 0.6-17 [57] with maximum likelihood estimation and standardization of both observed and latent variable covariances. Model fit was assessed based on common fit indices, i.e., Comparative Fit Index (CFI), Root Mean Squared Error of Approximation (RMSEA), Standardized Root Mean Square Residual (SRMR), and Tucker-Lewis Index (TLI). In accordance with previous literature and recommendations [58, 59], we considered the model to fit the data well when CFI and TLI  $\geq$  0.95, RMSEA < 0.06 and SRMR<0.08. The conducted CFA showed that a single latent factor model fits the data well (CFI = 0.99, RMSEA = 0.02, SRMR = 0.05, TLI = 0.99). Standardized parameter estimates for the model are presented in Table 6 and Fig.

Table 6	Standardized factor	loadings for	latent	constructs	from
confirma	atory factor analysis				

ltem	Standardized factor loading	SE	Z	p
1	0.10	0.09	1.05	0.295
2	-0.10	0.09	- 1.07	0.286
3	-0.14	0.09	-1.48	0.140
4	-0.76	0.08	-9.71	< 0.001
5	-0.83	0.08	- 10.99	< 0.001
6	-0.86	0.07	-11.64	< 0.001
7	-0.66	0.08	-8.10	< 0.001

The confirmatory factor analysis was conducted using the Maximum Likelihood estimation with the NLMINB optimization method and standardization of both observed and latent variable covariances. CFI=0.99, RMSEA=0.02, SRMR=0.05, TLI=0.99

3. Overall, the results support the validity of the measurement model and the hypothesized single-factor structure.

Despite the adequate fit, the internal consistency of the items was suboptimal (Cronbach's  $\alpha$ =0.69, McDonald's  $\omega$ =0.71). We therefore submitted the items of our stigma-related perceptions measure to an additional exploratory factor analysis (EFA) and re-analyzed our data based on the suggested splitting of items.

#### Exploratory factor analysis

The EFA was performed using the maximum likelihood extraction method and orthogonal rotation based on Varimax procedure. The analysis was conducted in R version 4.3.1 [56] using package psych 2.4.3 [71]. The Kaiser–Meyer–Olkin measure of sampling adequacy (KMO) and Bartlett's test of sphericity were used to ensure the conductibility of the EFA. The KMO resulted in a value of 0.8, which is higher than the minimum value of 0.5 [72], indicating that a sufficient amount of variance of variables could be explained by underlying factors. Additionally, Bartlett's test of sphericity reached statistical significance ( $\chi 2=252.91$ , p < 0.001), indicating that the data were suitable for factor analysis. Horn's parallel analysis (see Fig.

4) suggested extracting two factors. After extraction, items 2 and 3 were removed due to below-threshold communalities of lower than 0.30. We further removed items with factor loadings lower than 0.40, which is a common criterium for items to be clinically meaning-ful [73]. The remaining items were then assigned to the factor in which they had the highest loading. Thus, the EFA suggested averaging items 4 to 7 as one factor and maintaining item 1 as a second factor (see Table 7).



Fig. 3 Latent variable model showing relationships between items and perceived stigma





Fig. 4 Horn's parallel analysis

# Re-analysis of the data based on the two-factor solution

We re-analyzed the data using the mean score of items 4 to 7 (moderator W) and item 1 responses (moderator Z)

as two independent moderators in PROCESS version 4.2, with bootstrapped (5000 samples) bias-corrected 95% confidence interval (CI), heteroscedasticity correction

**Table 7** Factor structure of stigma-related perceptions of eating disorders in men measure by exploratory factor analysis

Item	Factor loadin	g	Communality
	Factor 1	Factor 2	
1	-0.17	0.57	0.35
2	0.08	0.15	0.03 <sup>a</sup>
3	0.10	0.35	0.13 <sup>a</sup>
4	0.78	-0.06	0.60
5	0.82	0.13	0.68
6	0.86	0.07	0.74
7	0.64	0.27	0.48

The extraction method was maximum likelihood extraction with orthogonal varimax rotation method. Bold type indicates the respective loadings of the items that were crucial for the assignment to the respective factor

<sup>a</sup> Item removed due to below-threshold communality and clinical meaningfulness

(HC3), and mean centering of predictor and moderator variables. As in the primary analysis, intention to seek help served as criterion Y, and the symptom severity measures as independent predictors X.

The association between the EDE-Q and help-seeking was primarily moderated by the score of items 4 to 7 (i.e., stigma-related perceptions for ED treatment), with item 1 (generally perceiving EDs as gender-stereotyped) having a reduced but non-significant effect. As in the primary analysis, no other moderations were detected (see Table 8). The pattern of conditional effects also remained comparable when only including the score of items 4 to 7 as moderator, without item 1. This pattern could suggest a more prominent impact of stigma-related perceptions that specifically concern ED treatment on help-seeking intentions compared to general ED-related stereotypes. However, given that factors with less than three items are generally weak

Table 8 Moderation analyses with two independent stigma-related perceptions of eating disorders in men categories

Relationship	Model st	statistics Pa		Path statistics			
	F	df1, df2	p	b	SE	p	95% CI [ <i>LL, UL</i> ]
EDE-Q	9.93	5, 126	< 0.001**				
Х→Ү				0.53	0.08	< 0.001**	0.38, 0.68
W→Y				-0.21	0.12	0.110	-0.45, 0.03
Z→Y				-0.20	0.10	0.038*	-0.39, 0.01
X*W→Y				-0.20	0.10	0.049*	-0.39, 0.01
X*Z→Y				-0.14	0.11	0.207	-0.34, 0.11
MDDI	3.99	5, 126	0.002**				
X→Y				0.05	0.01	< 0.001**	0.03, 0.08
W→Y				-0.39	0.19	0.041*	-0.78, -0.05
Z→Y				-0.15	0.11	0.158	-0.37, 0.06
X*W→Y				-0.01	0.01	0.539	-0.03, 0.02
X*Z→Y				-0.01	0.01	0.389	-0.04, 0.02
DOS	1.90	5, 126	0.098				
Х→Ү				0.07	0.02	0.006**	0.03, 0.12
W→Y				-0.26	0.16	0.106	-0.58, 0.04
Z→Y				-0.18	0.11	0.112	-0.41, 0.03
X*W→Y				-0.03	0.03	0.345	-0.09, 0.04
X*Z→Y				-0.01	0.03	0.688	-0.07, 0.04
EDY-Q	0.74	5, 112	0.595				
Х→Ү				0.02	0.13	0.886	-0.26, 0.26
W→Y				-0.30	0.19	0.149	-0.70, 0.04
Z→Y				-0.21	0.14	0.177	-0.48, 0.08
X*W→Y				-0.09	0.28	0.781	-0.67, 0.41
X*Z→Y				0.00	0.21	0.992	-0.39, 0.45

Values were computed with PROCESS version 4.2 model 2 using bootstrapping procedure with 5000 iterations, heteroscedasticity correction (HC3), and mean centering of predictor and moderator variables. X = respective symptom severity as independent variable; Y = intention to seek help as dependent variable; W = mean score of stigma-related perceptions of eating disorders in men items 4 to 7 as moderator; Z = stigma-related perceptions of eating disorders in men item 1 as moderator; EDE-Q = Eating Disorder Examination-Questionnaire; MDDI = Muscle Dysmorphic Disorder Inventory; DOS = Duesseldorf Orthorexia Scale; EDY-Q = Eating Disorders in Youth-Questionnaire; CI = confidence interval; LL = lower limit; UL = upper limit

\*p<0.05. \*\*p<0.01

and unstable [67], and that a single-factor solution, as indicated by the reported CFA, fits the data well, we caution against further interpretation and opted to aggregate the items on perceived stigma into a single mean score for further analyses.

#### Abbreviations

ARFID	Avoidant/restrictive eating disorder
CFA	Confirmatory factor analysis
CFI	Comparative fit index
DOS	Duesseldorf orthorexia scale
ED	Eating disorder
EDE-Q	Eating disorder examination-questionnaire
EDY-Q	Eating disorders in youth-questionnaire
EFA	Exploratory factor analysis
КМО	Kaiser–Meyer–Olkin measure of sampling adequacy
Μ	Mean
Max	Maximum value
MDDI	Muscle dysmorphic disorder inventory
Min	Minimum value
RMSEA	Root mean squared error of approximation
SD	Standard deviation
SOCQ-ED	Stages of change questionnaire for eating disorders
SRMR	Standardized root mean square residual
TLI	Tucker–Lewis index

#### Acknowledgements

The present work was performed by Martin S. Lehe in partial fulfillment of the requirements for obtaining the degree "Dr. phil."

#### Author contributions

MSL: conceptualization, formal data analysis, visualization, writing and revision of initial draft GH: conceptualization, provided supervision in data analysis and writing, reviewed and supported editing of manuscript SSL: conceptualization, reviewed and supported editing of manuscript GP: conceptualization, provided resources and supervision, reviewed and supported editing of manuscript.

#### Funding

Open Access funding enabled and organized by Projekt DEAL. The data derived from a project funded by the German Society for Eating Disorders (DGESS).

#### Availability of data and materials

The dataset and materials can be obtained from the corresponding author upon request.

#### Declarations

#### Ethics approval and consent to participate

The data collection obtained ethical approval under reference AZ 2022-910 on April 21st, 2022, and was registered at https://aspredicted.org/5T3\_NH5.

#### **Consent for publication**

All authors consented to publication. Details may be found in the attached document.

#### Competing interests

The authors declare no competing interests.

#### Author details

<sup>1</sup>University Clinic for Psychosomatic Medicine and Psychotherapy, Medical Faculty, Campus East-Westphalia, Ruhr-University Bochum, Virchowstr. 65, 32312 Lübbecke, Germany. <sup>2</sup>Department of Clinical Psychology and Psychotherapy, Otto-Friedrich-University of Bamberg, Markusplatz 3, 96047 Bamberg, Germany.

#### Received: 27 February 2024 Accepted: 11 November 2024 Published online: 05 December 2024

#### References

- Hay PJ, Mond J. How to 'count the cost' and measure burden? A review of health-related quality of life in people with eating disorders. J Ment Health. 2005;14(6):539–52.
- van Hoeken D, Hoek HW. Review of the burden of eating disorders: mortality, disability, costs, quality of life, and family burden. Curr Opin Psychiatry. 2020;33(6):521–7.
- Padierna A, Martín J, Aguirre U, González N, Muñoz P, Quintana JM. Burden of caregiving amongst family caregivers of patients with eating disorders. Soc Psychiatry Psychiatr Epidemiol. 2013;48(1):151–61.
- Stefanini MC, Troiani MR, Caselli M, Dirindelli P, Lucarelli S, Caini S, et al. Living with someone with an eating disorder: factors affecting the caregivers' burden. Eat Weight Disord. 2019;24(6):1209–14.
- Streatfeild J, Hickson J, Austin SB, Hutcheson R, Kandel JS, Lampert JG, et al. Social and economic cost of eating disorders in the United States: Evidence to inform policy action. Int J Eat Disord. 2021;54(5):851–68.
- Lavender JM, Brown TA, Murray SB. Men, muscles, and eating disorders: an overview of traditional and muscularity-oriented disordered eating. Curr Psychiatry Rep. 2017;19(6):32.
- Qian J, Wu Y, Liu F, Zhu Y, Jin H, Zhang H, et al. An update on the prevalence of eating disorders in the general population: a systematic review and meta-analysis. Eat Weight Disord. 2022;27(2):415–28.
- Santomauro DF, Melen S, Mitchison D, Vos T, Whiteford H, Ferrari AJ. The hidden burden of eating disorders: an extension of estimates from the Global Burden of Disease Study 2019. Lancet Psychiatry. 2021;8(4):320–8.
- GBD Mental Disorders Collaborators Group. Global, regional, and national burden of 12 mental disorders in 204 countries and territories, 1990–2019: a systematic analysis for the Global Burden of Disease Study 2019. Lancet Psychiatry. 2022;9(2):137–50.
- Galmiche M, Déchelotte P, Lambert G, Tavolacci MP. Prevalence of eating disorders over the 2000–2018 period: a systematic literature review. Am J Clin Nutr. 2019;109(5):1402–13.
- Gull WW. Anorexia nervosa (apepsia hysterica, anorexia hysterica). Trans Clin Soc Lond. 1874;7:22–8.
- 12. Lasègue EC. De l'anorexie hystérique. Arch Gén Méd. 1873;1:385-403.
- Morton R. A treatise of consumptions. London: Sam Smith and Benjamin Walford; 2012. p. 1694.
- Murray SB, Nagata JM, Griffiths S, Calzo JP, Brown TA, Mitchison D, et al. The enigma of male eating disorders: a critical review and synthesis. Clin Psychol Rev. 2017;57:1–11.
- Brown TA, Keel PK. Eating disorders in boys and men. Annu Rev Clin Psychol. 2023;19(1):177–205.
- Gorrell S, Murray SB. Eating disorders in males. Child Adolesc Psychiatr Clin N Am. 2019;28(4):641–51.
- 17. Nagata JM, Ganson KT, Murray SB. Eating disorders in adolescent boys and young men: an update. Curr Opin Pediatr. 2020;32(4):476–81.
- Mitchison D, Mond J, Griffiths S, Hay P, Nagata JM, Bussey K, et al. Prevalence of muscle dysmorphia in adolescents: findings from the EveryBODY study. Psychol Med. 2022;52(14):3142–9.
- White M, Berry R, Rodgers RF. Body image and body change behaviors associated with orthorexia symptoms in males. Body Image. 2020;1(34):46–50.
- Eddy KT, Thomas JJ, Hastings E, Edkins K, Lamont E, Nevins CM, et al. Prevalence of DSM-5 avoidant/restrictive food intake disorder in a pediatric gastroenterology healthcare network. Int J Eat Disord. 2015;48(5):464–70.
- Strother E, Lemberg R, Stanford SC, Turberville D. Eating disorders in men: underdiagnosed, undertreated, and misunderstood. Eat Disord. 2012;20(5):346–55.
- Smith GD, Hebdon M. Mental health help-seeking behaviour in men. J Adv Nurs. 2024;80(3):851–3.
- 23. Vickery A. Men's help-seeking for distress: navigating varied pathways and practices. Front Sociol. 2021;26(6): 724843.
- 24. Prentice DA, Carranza E. What women and men should be, shouldn't be, are allowed to be, and don't have to be: the contents of prescriptive gender stereotypes. Psychol Women Q. 2002;26(4):269–81.
- Moss-Racusin CA, Phelan JE, Rudman LA. When men break the gender rules: status incongruity and backlash against modest men. Psychol Men Masc. 2010;11(2):140–51.
- Vandello JA, Bosson JK. Hard won and easily lost: a review and synthesis of theory and research on precarious manhood. Psychol Men Masc. 2013;14(2):101–13.

- Vandello Joseph A, Bosson JK, Lawler JR. Precarious manhood and men's health disparities. In: Griffith DM, Bruce MA, Thorpe RJ Jr, editors. Men's health equity: a handbook. Milton Park: Routledge/Taylor & Francis Group; 2019. p. 27–41.
- Richardson C, Paslakis G. Men's experiences of eating disorder treatment: a qualitative systematic review of men-only studies. J Psychiatr Ment Health Nurs. 2021;28(2):237–50.
- Bomben R, Robertson N, Allan S. Barriers to help-seeking for eating disorders in men: a mixed-methods systematic review. Psychol Men Masc. 2022;23(2):183–96.
- Grillot CL, Keel PK. Barriers to seeking treatment for eating disorders: the role of self-recognition in understanding gender disparities in who seeks help. Int J Eat Disord. 2018;51(11):1285–9.
- Griffiths S, Mond JM, Murray SB, Touyz S. The prevalence and adverse associations of stigmatization in people with eating disorders. Int J Eat Disord. 2014;48(6):767–74.
- Griffiths S, Mond JM, Li Z, Gunatilake S, Murray SB, Sheffield J, et al. Selfstigma of seeking treatment and being male predict an increased likelihood of having an undiagnosed eating disorder. Int J Eat Disord. 2015;48(6):775–8.
- Hackler AH, Vogel DL, Wade NG. Attitudes toward seeking professional help for an eating disorder: the role of stigma and anticipated outcomes. J Couns Dev. 2010;88(4):424–31.
- 34. McArdle S, Moore P. Exploring indices of disordered eating literacy in male and female athletes: knowledge gaps and lay beliefs as predictors of attitudes toward help-seeking. J Clin Sport Psychol. 2013;7(4):275–92.
- Ali K, Fassnacht DB, Farrer L, Rieger E, Feldhege J, Moessner M, et al. What prevents young adults from seeking help? Barriers toward help-seeking for eating disorder symptomatology. Int J Eat Disord. 2020;53(6):894–906.
- Sansfaçon J, Booij L, Gauvin L, Fletcher É, Islam F, Israël M, et al. Pretreatment motivation and therapy outcomes in eating disorders: a systematic review and meta-analysis. Int J Eat Disord. 2020;53(12):1879–900.
- Tucker JR, Hammer JH, Vogel DL, Bitman RL, Wade NG, Maier EJ. Disentangling self-stigma: are mental illness and help-seeking self-stigmas different? J Couns Psychol. 2013;60(4):520–31.
- Byrne SE, Basten CJ, McAloon J. The development of disordered eating in male adolescents: a systematic review of prospective longitudinal studies. Adolescent Res Rev. 2023;9(2):227–52.
- Laskowski NM, Halbeisen G, Braks K, Huber TJ, Paslakis G. Factor structure of the eating disorder examination-questionnaire (EDE-Q) in adult men with eating disorders. J Eat Disord. 2023;11(1):34.
- Núñez-Navarro A, Agüera Z, Krug I, Jiménez-Murcia S, Sánchez I, Araguz N, et al. Do men with eating disorders differ from women in clinics, psychopathology and personality? Eur Eat Disord Rev. 2012;20(1):23–31.
- 41. Pearl RL, Austin SB, Jebeile H, D'Adamo L, Wilfley DE. Opportunities to advance research, intervention, and policy on stigma, eating disorders, and body image. J Eat Disord. 2023;11(1):132.
- 42. Eschrich RL, Halbeisen G, Steins-Löber S, Timmesfeld N, Paslakis G. Investigating the structure of disordered eating symptoms in adult men: a network analysis. Eur Eat Disord Rev 2024.
- Faul F, Erdfelder E, Lang AG, Buchner A. G\*Power 3: a flexible statistical power analysis program for the social, behavioral, and biomedical sciences. Behav Res Methods. 2007;39(2):175–91.
- Cohen J. Statistical power analysis for the behavioral sciences. 2nd ed. Hillsdale: L. Erlbaum Associates; 1988. p. 567.
- 45. Clement S, Schauman O, Graham T, Maggioni F, Evans-Lacko S, Bezborodovs N, et al. What is the impact of mental health-related stigma on help-seeking? A systematic review of quantitative and qualitative studies. Psychol Med. 2015;45(1):11–27.
- De Leeuw JR, Gilbert RA, Luchterhandt B, jsPsych: enabling an open-source collaborative ecosystem of behavioral experiments. JOSS. 2023;8(85):5351.
- Richter F, Strauss B, Braehler E, Altmann U, Berger U. Psychometric properties of a short version of the eating attitudes test (EAT-8) in a German representative sample. Eat Behav. 2016;21:198–204.
- 48. Fairburn CG, Beglin SJ. Eating disorder examination questionnaire. 1994.
- Hilbert A, Tuschen-Caffier B. Eating disorder examination questionnaire. 2. Auflage. Tübingen: Dgvt-Verlag; 2016.
- Zeeck A, Welter V, Alatas H, Hildebrandt T, Lahmann C, Hartmann A. Muscle dysmorphic disorder inventory (MDDI): validation of a German version with a focus on gender. PLoS ONE. 2018;13(11):e0207535.

- Barthels F, Meyer F, Pietrowsky R. Die Düsseldorfer Orthorexie Skala-Konstruktion und Evaluation eines Fragebogens zur Erfassung ortho-rektischen Ernährungsverhaltens. Z Klin Psychol Psychother. 2015;44(2):97–105.
- 52. Hilbert A, van Dyck Z. Eating disorders in youth-questionnaire (Deutsche Version). Leipzig: Universität Leipzig; 2016.
- 53. Hilbert A, Zenger M, Eichler J, Brähler E. Psychometric evaluation of the eating disorders in youth-questionnaire when used in adults: prevalence estimates for symptoms of avoidant/restrictive food intake disorder and population norms. Int J Eat Disorders. 2021;54(3):399–408.
- Von Brachel R, Hötzel K, Schloßmacher L, Hechler T, Kosfelder J, Rieger E, et al. Entwicklung und Validierung einer deutschsprachigen Skala zur Erfassung der Veränderungsmotivation bei Essstörungen – the stages of change questionnaire for eating disorders (SOCQ-ED). Psychother Psych Med. 2012;62(12):450–5.
- Prochaska JO, Velicer WF. The transtheoretical model of health behavior change. Am J Health Promot. 1997;12(1):38–48.
- 56. R Core Team. R: a language and environment for statistical computing. Vienna: R Foundation for Statistical Computing; 2021.
- Rosseel Y. Iavaan: an R Package for Structural Equation Modeling. J Stat Softw. 2012;48(2):1–36.
- Hu LT, Bentler PM. Cutoff criteria for fit indexes in covariance structure analysis: Conventional criteria versus new alternatives. Struct Equ Model A Multidiscip J. 1999;6(1):1–55.
- Schreiber JB, Nora A, Stage FK, Barlow EA, King J. Reporting structural equation modeling and confirmatory factor analysis results: a review. J Educ Res. 2006;99(6):323–38.
- 60. Hayes AF. Introduction to mediation, moderation, and conditional process analysis: a regression-based approach. 3rd ed. New York, London: The Guilford Press; 2022. p. 732.
- 61. IBM Corp. IBM SPSS statistics for windows 28.0. 2022.
- 62. Lin H. Probing two-way moderation effects: a review of software to easily plot johnson-neyman figures. Struct Equ Modeling. 2020;27(3):494–502.
- Wickham H. ggplot2: elegant graphics for data analysis. New York: Springer; 2016.
- Hemmerich W. StatistikGuru: Johnson-Neyman-Plots erstellen [Internet]. 2024. Available from: https://statistikguru.de/rechner/johnson-neymanplots-erstellen.html
- Yousaf O, Grunfeld EA, Hunter MS. A systematic review of the factors associated with delays in medical and psychological help-seeking among men. Health Psychol Rev. 2015;9(2):264–76.
- Halbeisen G, Brandt G, Paslakis G. A plea for diversity in eating disorders research. Front Psychiatry. 2022;13: 820043.
- Costello AB, Osborne J. Best practices in exploratory factor analysis: four recommendations for getting the most from your analysis. Pract Assess Res Eval 2005;10(1):1–9. https://doi.org/10.7275/jyj1-4868
- Fox AB, Earnshaw VA, Taverna EC, Vogt D. Conceptualizing and measuring mental illness stigma: the mental illness stigma framework and critical review of measures. Stigma Health. 2018;3(4):348–76.
- Innes NT, Clough BA, Casey LM. Assessing treatment barriers in eating disorders: a systematic review. Eat Disord. 2017;25(1):1–21.
- Klik KA, Williams SL, Reynolds KJ. Toward understanding mental illness stigma and help-seeking: a social identity perspective. Soc Sci Med. 2019;1(222):35–43.
- Revelle W. psych: procedures for psychological, psychometric, and personality research. Evanston, Illinois: Northwestern University; 2017. Available from: https://CRAN.R-project.org/package=psych
- 72. Dziuban CD, Shirkey EC. When is a correlation matrix appropriate for factor analysis? Some decision rules. Psychol Bull. 1974;81(6):358–61.
- Beavers AS, Lounsbury JW, Richards JK, Huck SW, Skolits GJ, Esquivel SL. Practical considerations for using exploratory factor analysis in educational research.

### **Publisher's Note**

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.